Author's response to reviews

Title: Health Facility Barriers to HIV Linkage and Retention in Western Kenya

Authors:

Juddy W Wachira (wachirajuddy@gmail.com)
Violet Naanyu (vnaanyu@gmail.com)
Beatrice Koech (beatyjelagat@yahoo.com)
Jacqueline Akinyi (jackiezulu2007@yahoo.com)
Regina Kamene (rkamene70@gmail.com)
Becky Genberg (becky_genberg@brown.edu)
Samson Ndege (ndegekip@yahoo.com)
Abraham Siika (amsiika@africaonline.co.ke)
Sylvester Kimaiyo (skimaiyo@ampathplus.or.ke)
Paula Braitstein (pbraitstein@yahoo.com)

Version: 2
Date: 23 July 2014

Author's response to reviews:

23/7/2014

To:
The Editor
BMC Health Service Research Journal

Dear Sir/Madam

REF: Health Facility Barriers to HIV Linkage and Retention in Western Kenya

Thank you the recommended revisions on our manuscript entitled: “Health Facility Barriers to HIV Linkage and Retention in Western Kenya”.

We have made the following revisions:

Title page: The title page now includes the email addresses for all the co-investigators.

Manuscript body: We have highlighted the different sections as recommended by the journal. We have also included a section on human subject protection.

We look forward to the consideration of this manuscript in your journal.

Sincerely,
Corresponding author:
Juddy Wachira, PhD, MPH,
Moi University,
Department of Behavioral Sciences,
P.O Box 4606,
Eldoret-30100,
Kenya.
Tel: +254705242450
Email: wachirajuddy@gmail.com